Author's response to reviews

Title: Impact of a program to prevent incivility towards and assault of healthcare staff in an ophtalmological emergency unit. Study protocol for the PREVURGO On/Off trial.

Authors:

Sandrine Touzet (sandrine.touzet@chu-lyon.fr)
Pierre-Loic Cornut (dr.cornut@polevision.fr)
Jean-Baptiste Fassier (jean-baptiste.fassier@chu-lyon.fr)
Marie-Annick Le Pogam (marie-annick.le-pogam@chuv.ch)
Carole Burillon (carole.burillon@chu-lyon.fr)
Antoine Duclos (antoine.duclos@chu-lyon.fr)

Version: 3
Date: 12 May 2014

Author's response to reviews:

The authors have NO other articles relating to the protocol published or under consideration.